Author's response to reviews

Title: The optimal starting time of postoperative intraperitoneal mitomycin-C therapy with preserved intestinal wound healing

Authors:

Ali Uzunkoy (aliuzunkoy@yahoo.com)
Cengiz Bolukbas (drcengizbolukbas@hotmail.com)
Mehmet Horoz (mehmethoroz@yahoo.com)
Fusun F Bolukbas (fusunbol@yahoo.ca)
Abdurrahim Kocyigit (kocyigit@harran.edu.tr)

Version: 7 Date: 5 March 2005

Author's response to reviews:

Dear Editor, March 05, 2005
Thanks for your attention that you have paid to our manuscript entitled "The optimal starting time of postoperative intraperitoneal mitomycin-C therapy with preserved intestinal wound healing (MS: 4076810250841512)". We revised the manuscript according to your suggestion and point-to-point description of revisions made are listed below:
1- We included include the country at the end of all the affiliation details.
2- We elaborated the Background section to place the study in the context of the current knowledge in the field and we also updated the abstract details on the submission page.
3-We changed the author contributions section to 'Authors' contributions'.
4- Some language corrections were made by a researcher who is native in English.
With Best Regards,
Mehmet HOROZ, MD.